Author's response to reviews

Title: Primary prevention of diabetes mellitus type 2 and cardiovascular diseases using a cognitive behavior program aimed at lifestyle changes in people at risk: design of a randomized controlled trial

Authors:

Jeroen Lakerveld (j.lakerveld@vumc.nl)
Sandra D.M. Bot (s.bot@vumc.nl)
Marijke J. Chinapaw (m.chinapaw@vumc.nl)
Maurits W. van Tulder (mw.vantulder@vumc.nl)
Patricia van Oppen (pvanoppen@ggzba.nl)
Jacqueline M. Dekker (jm.dekker@vumc.nl)
Giel Nijpels (g.nijpels@vumc.nl)

Version: 2 Date: 29 May 2008

Author's response to reviews:

Dear Editor,

We have recently been submitting our article to BMC Medicine. However, according to Robin Cassady-Cain (assistant editor of BMC Medicine), our article would be ideally suited to the scope and readership of BMC Endocrine Disorders. We hope that you agree, and that you will consider publication of this study in BMC Endocrine Disorders.

Sincerely,
Jeroen Lakerveld,
corresponding author